N. CHOHAN/E. VEYRUNES/01-August-05

Week 31  starting 01 Aug 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  01 August 05 : 8/1/2005 3:40:33 PM
	6
	A1
	MBAL2079
	Prep 5 Cool Down
	290.98

	4
	A2
	MBBR3381
	Prep 5 Cool Down
	2.35

	10
	B1
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	298.04

	12
	B2
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	298.49

	9
	C1
	MBBR3311
	ICS 1 WP04 HV Test Warm
	297.41

	3
	C2
	MBAL1227
	PT 12 Warm Up
	205.88

	8
	D1
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	297.86

	11
	D2
	SSS607
	ICS 4 Final connection
	297.77

	5
	E1
	MBAL1178
	Prep 3 CDW1 HV
	299.11

	2
	E2
	MBAR3368
	PT 13.2 Resist. Meas.
	302.52

	7
	F1
	MBAR3375
	ICS 3 Traveller after warm test
	297.15

	1
	F2
	MBBR3379
	PT 13.2 Resist. Meas.
	299.65


   
Coordinator this week: Nicolay Smirnov; Federico Chiusano MTM support 16-3602

16 complete Magnet Tests cycles
12 dipoles ( 3 with MM )+ 4 SSS(without MM & SSW )
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
General Points :

· Special S4 (Jackie)

· Rotary Switch for Corrector powering (awaiting Giorgio?).

· Cluster F “INTERCOM Box” repair (awaiting Giorgio?).
Some details  for the  last  month’s (July 05)  tests output:

1. Early July (29June—10July) SSS203 had SSW done at warm,  cold & warm again – a time consuming exercise for ssw systems tests; + it had full shafts m’ments as well
2. 2051  :[ tested 9 days Between  15 July-24 July] 

FOUR  4.4K SSL Tests were  done (= ~12 hrs).

:’ Optical tests’ Effect of Field on light tests (floor level)   6 hrs
i. -Extra HV Vclamp vs dipole.

3. Fatigue Test SSS217 on D1 ( extra ~62 Hours). Total ~ 9 Days

4. 3336 : Warm and Cold SSW in a dipole  ( > 38 Hours).  Weekend  magnet cold  kept to do cold SSW on Monday , config files, Survey issues,  experts missing etc

5. Silver Shoe Repair on D2 before bringing in Q4 (sss607)
 So Bench D2 empty for this  repair work over weekend 23rd july + some weekdays 25/26 july
6. Over the weekend of 30/31 July no further work could be be done on D2 because connection by experts was stopped on Fri evening 29th July  & would only resume on Monday 1 Aug

7. Benches B1 & B2 need conversion to Dipole &/or Connection cryostat because no more SSS ‘s available. Hence Benches remained empty substantially since weekend of 30/31July05  
8. New Type of correction connections on SSS meant lot of problems on ICS side & several re-openings after connection( c17 D19 , E13 ,F11  INSTEAD of previous) & much time Lost in connection
9. In July, the arc-SSS [Wait+cooling time]  went up to avrg. 31 hrs instead of ~ 20 hr avrg in previous months due to SSS199 on B1 in cohabitation with B2 and SSS036  having many HV tests during cooldown(during working hrs Mon-Fri), blocking any useful progress on B1 bench SSS199  in cooling down faster even though process was launched 

	Magnets under Test from 24-07-2005 to 30-07-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for

	1
	TBA1
	MBAR3371 
	[image: image7.png]



	25-07-2005 
	35 h.
	31 h.
	30 h.
	18 h.
	8 h.
	30-07-2005 
	122 h.
	Stripping

	2
	TBA1
	MBBL1233 
	[image: image8.png]



	20-07-2005 
	37 h.
	24 h.
	18 h.
	14 h.
	21 h.
	25-07-2005 
	114 h.
	Stripping

	3
	TBA2
	MBAL2051(2) 
	[image: image9.png]



	15-07-2005 
	49 h.
	67 h.
	68 h.
	19 h.
	20 h.
	24-07-2005 
	223 h.
	Stand By

	4
	TBA2
	MBAL2170 
	[image: image10.png]



	24-07-2005 
	30 h.
	21 h.
	37 h.
	23 h.
	18 h.
	30-07-2005 
	129 h.
	Stripping

	5
	TBB1
	SSS173 
	[image: image11.png]



	23-07-2005 
	38 h.
	14 h.
	16 h.
	8 h.
	14 h.
	26-07-2005 
	90 h.
	Stripping

	6
	TBB2
	SSS036(5) 
	[image: image12.png]



	17-07-2005 
	11 h.
	92 h.
	34 h.
	11 h.
	26 h.
	25-07-2005 
	174 h.
	Stand By

	7
	TBB2
	SSS195 
	[image: image13.png]



	25-07-2005 
	48 h.
	14 h.
	20 h.
	9 h.
	10 h.
	30-07-2005 
	101 h.
	Stand By

	8
	TBC1
	MBBR3360 
	[image: image14.png]



	22-07-2005 
	28 h.
	21 h.
	14 h.
	12 h.
	18 h.
	26-07-2005 
	93 h.
	Stripping

	9
	TBC2
	MBBL3121 
	[image: image15.png]



	23-07-2005 
	41 h.
	25 h.
	26 h.
	15 h.
	12 h.
	29-07-2005 
	119 h.
	Stripping

	10
	TBD1
	SSS152 
	[image: image16.png]



	22-07-2005 
	40 h.
	15 h.
	19 h.
	8 h.
	22 h.
	26-07-2005 
	104 h.
	Stand By

	11
	TBE1
	MBBL3115 
	[image: image17.png]



	22-07-2005 
	35 h.
	21 h.
	21 h.
	13 h.
	27 h.
	27-07-2005 
	117 h.
	Stripping

	12
	TBE2
	MBAL1207 
	[image: image18.png]



	24-07-2005 
	36 h.
	22 h.
	16 h.
	15 h.
	10 h.
	28-07-2005 
	99 h.
	Stripping

	13
	TBE2
	MBAL1243 
	[image: image19.png]



	18-07-2005 
	25 h.
	43 h.
	37 h.
	16 h.
	13 h.
	24-07-2005 
	134 h.
	Stripping

	14
	TBF1
	MBBL1195 
	[image: image20.png]



	26-07-2005 
	33 h.
	21 h.
	17 h.
	22 h.
	10 h.
	30-07-2005 
	103 h.
	Stripping

	15
	TBF1
	MBBR3336 
	[image: image21.png]



	18-07-2005 
	62 h.
	14 h.
	74 h.
	15 h.
	23 h.
	26-07-2005 
	188 h.
	Stripping

	16
	TBF2
	MBAR3367 
	[image: image22.png]



	22-07-2005 
	63 h.
	23 h.
	14 h.
	14 h.
	31 h.
	28-07-2005 
	145 h.
	Stripping

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10

	1
	MBAR3371
	12850 A.
(D1-U-S13)
	12850 A.
[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL1233
	12133.5 A.
(D1_L)
	12527.2 A.
(D2_L)
	12850 A.
(D1_L)
	[image: image24.png]



	-
	-
	-
	-
	-
	-

	3
	MBAL2051
	11995 A.
(D1-MRB)
	12592 A.
(CF-S04-A)
	12640 A.
(D1-U-MRB)
	12639.2 A.
(D1-U)
	12645 A.
(D1-U)
	12649.4 A.
(D1_U)
	[image: image25.png]



	-
	-
	-

	4
	MBAL2170
	11565 A.
(D1-L-S01)
	12228 A.
(D2-U-S01)
	12443 A.
(D1-U-S01)
	12636 A.
(D2_L_S01)
	[image: image26.png]



	-
	-
	-
	-
	-

	5
	SSS173
	12130.3 A.
(QI--12)
	12329.9 A.
(QI-34)
	12534.7 A.
(QE_34)
	[image: image27.png]



	-
	-
	-
	-
	-
	-

	6
	SSS036
	12281 A.
(QE-34)
	12818 A.
(QI-12)
	12800 A.
(QI-12)
	[image: image28.png]



	-
	-
	-
	-
	-
	-

	7
	SSS195
	11103 A.
(QI-12)
	12108 A.
(QI-12)
	12761 A.
(QI-12)
	[image: image29.png]



	-
	-
	-
	-
	-
	-

	8
	MBBR3360
	12136 A.
(D1-U)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL3121
	9596 A.
(D1_U)
	12050.8 A.
(D1-U)
	11746 A.
(D1_L)
	12651 A.
(D1-U)
	12829.8 A.
(D2-U)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-

	10
	SSS152
	12129 A.
(QI_34)
	12291 A.
(QI_12)
	12548 A.
(QE-12)
	[image: image32.png]



	-
	-
	-
	-
	-
	-

	11
	MBBL3115
	10691.1 A.
(D1_L)
	11786 A.
(D1-U)
	12351 A.
(D2-L)
	12850 A.
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	-
	-
	-
	-
	-
	-

	12
	MBAL1207
	12536 A.
(D1-L)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAL1243
	11703 A.
(D1-L)
	11842.4 A.
(D1_L)
	12133 A.
(D2-L)
	12297 A.
(D2-L)
	12437 A.
(D1-U)
	12588 A.
(D1_L)
	12667.2 A.
(D1-L)
	12824.6 A.
(D1-L)
	12850 A.
[image: image35.jpg]



	-

	14
	MBBL1195
	12342 A.
(D2-L)
	12867 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR3336
	10515.2 A.
(D2-U)
	11971 A.
(D2_U)
	12193 A.
(D2-U)
	12439.5 A.
(D1_U_S11)
	12745 A.
(D1_L_S02)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-

	16
	MBAR3367
	9603.9 A.
(D2_L)
	12323 A.
(D2 - L)
	12573.3 A.
(D2-U)
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	-
	-
	-
	-
	-
	-
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	Date
	Bench
	Problem
	Type
	Application/
soft
	Action/advise

	Tuesday _26_A
	TBA1

	ICS informed that resistance R2 of short anticryostat (22)in Ap.2 not OK.
	MTM
	
	Salem is being informed.

	Tuesday _26_A
	TBC2

	Problem with the polarity switch. Difficult to change from right to left
	MTM
	
	AB/PO has solved the problem.

	Tuesday _26_A
	TBD1

	Cryo problem: Compressor problem, not able to pump.
	ACR
	
	After 2 hours Cryo OK.

	Tuesday _26_N
	TBF2
TBC2


	Water Leak (Mark with a brown tape for F2, TCR has been called for C2 according Richard indication) 
	MTM
	
	Leak always present on Monday 01 August for C2, Mechanics people of SM18 has been informed.

	Wednesday _27_A
	TBF2

	Federico called, concerning quench currents on 3rd and 4th training of MBBR3336. voltmeter values show 15 amps higher than HF (AQA) values. He asks us to modify CDPT file, via SMTMS, to show the AQA current.

	MTM
	
	Resolved for next quench, may be a offset in voltmeter.

	Wednesday _27_N
	TBB2

	PC fault. We can't launch PT4. On PC rack downstairs, PC FAST is active on interlock module and not resetable (even in local).
We will inform equipment support around 07h00 but suspect PO crew has to come.
	OP
	
	Stop button was press after PO intervention.

	Thursday _28_M

Saturday_30_M
	TBA1

TBE1


	MBAR3371 SPA at 1406 during PT 5.1 provoked quench at 1500A. Cryo interlock not OK. water condensation on current leads

	ACR/

MTM
	
	Fun used to reduce the condensation.

	Thursday_28_N
Saturday_30_M
	TBB1

	PROBLEM WITH HV RACK (XMT.MY95). NO COMMUNICATION BETWEEN PLCSE AND PLC HV. PROBLEM MAY BE WITH THE HV RACK AND NO SPARE HV RACK IS AVAILABLE SO TEST HAS BEEN POSTPONED FOR THE TIME BEING. (SSS173)-Only one Mobile Rack(XMT.MY93) is available for HV. After F1, test has been launched on B1.
	MTM
	
	Richard Situation ?

	Sunday _31_A
	TBA2

	Magnet detrained in the warm-up quench (3rd Ramp). Test coordinator advised to perform another training quench (4th Ramp).
	MTM
	
	Magnet reach ULTIMATE 4th ramp.

	Sunday _31_N 

	TBC2

	De-Excitation. The Power application did not stop at 12000A, it continued ramping to 12850 (Good magnet!) The configuration was correct in Tema, but the Target Current was 12850 in the Power application.

	MTM
	
	?


Green Connect + Pump down /Blue :  Wait + Cool down	Ivory: Tests at Cold (36 hr)Red :  Wait + Warm up (12 hr)	  Dark Green: Disconnect (8 hr)











